STA213 Fall 1999

Introduction to Statistical Methods

Instructor : Jaeyong Lee

email : jaeyong@stat.duke.edu office : 211E Old Chem tel : 684-4447

T.A. : Maria DeIorio

email : maria@stat.duke.edu office : 221 Old Chem tel : 684-4558

Office Hours : Monday and Wednesday 3:35-4:35

or by appointment

Text : Morris H. Degroot, Probability and Statistics, second edition, Addison-Wesley, 1989.

Homework : Bi-weekly (no late homework is accepted unless your homework disappeared during earthquake, hurricane or flood, etc.)

Grading : Midterm (30%) Final Exam (40%) Homework (30%)

Topics : The schedule could change during the course of the semester.

Weeks : Topics (chapters) 1-3 : Basic Probability (1,2,3) 4-5 : Expectation (4) 6-8 : Special Distributions (5,7) and Midterm 9-11 : Estimation (6,7) 12-14 : Testing (8)